MA194 { Using Unix

Instructor: Dan Chaney (chaney@ms.uky.edu) Office hours: Mon 2-3, 606-257-2150

Class #12 - Tuesday, February 20, 1996 – Week 6

In-class hotline: (606) 323–3681 Community Colleges: Tues 9:00-9:50 & 10:00-10:50 PM EST UK/Lexington: Wed & Fri 9:00 PM EST, TCI Channel 16

URL: http://www.ms.uky.edu/~ma194

Mailing list: ma194-class@ms.uky.edu To join the mailing list, send a note to ma194@ms.uky.edu

Tonight’s Topics – Command redirection (Conclusion) – Homework 5 solutions/discussion – Special characters (meta-characters)

Shell meta-characters

< Redirect command input from a file
> Redirect command output to a file
>> Append command output to a file

* Match any string in filenames
? Match any single character in a filename
; Separate commands on a command line
& Background a command
’xxx’ Literally, the string xxx
\X Literally, the character X
$var Contents of the ’var’ environment variable
$* Last token on previous command